Author’s response to reviews

Title: A combination of left ventricular noncompaction and double orifice mitral valve

Authors:

Xing-Xiang Wang (wangxx0571@yahoo.com.cn)
Ze-Zhou Song (zezhou_song@126.com)

Version: 2 Date: 1 March 2009

Author’s response to reviews:

Dear Editor:
Thank you very much for your kind letter along with your and the reviewers’ comments. I greatly appreciate you and the reviewers for your support of my manuscript (1226767982257226) entitled “A combination of left ventricular noncompaction and double orifice mitral valve” to be published in “Cardiovascular ultrasound”. According to the suggestions and comments given by you and the reviewers I have revised the manuscript. My point-by-point response to your and the reviewers’ comments attached below. I would appreciate you very much if you let us know your decision at your earliest convenience. Wish this work could appear in “Cardiovascular ultrasound” soon. Thank you!

REVIEWER Comments:

Comments 1. On pg. 5 delete the following sentence, which is inappropriate: "In addition, the question must be posed as to whether coronary microvascular dysfunction and ventricular noncompaction are both the consequence of an additional undetermined cause?"

Answer: The above-mentioned sentence has been changed.

Comments 2. It would important that authors do upload more clips (there is no space limit).

Answer: I agree with the reviewer, however, I think that the provided figures are enough to reveal the case report and the rest figures of the case are cumbersome for publication.

Comments 3. Apparently, these very same authors seem to have quite a number of NVM and they may think of collect all these cases in the same manuscript.

Answer: I agree with the reviewer, however, the case of combination of NVM and double orifice mitral valve is the only one that I ever managed. I will collect all analogous cases in the same manuscript in the future submission.

Best guards

Xing-Xiang Wang and Ze-Zhou Song